Author's response to reviews

Title: Reducing bottlenecks: professionals' and adolescents' experiences with transitional care delivery

Authors:

Anna P Nieboer (nieboer@bmg.eur.nl)
Jane M Cramm (cramm@bmg.eur.nl)
Henk Sonneveld (mailto.hms@gmail.com)
Marij Roebroeck (m.roebroeck@erasmusmc.nl)
Anne L Staa van (vanstaa@bmg.eur.nl)
Mathilde MH Strating (strating@bmg.eur.nl)

Version: 2 Date: 17 July 2013

Author's response to reviews:

Dear Editor,

As requested I explained in the Methods section that informed consent was obtained from adolescents and parents. In addition, I changed the abstract according to BMC Health Services Research guidelines. I removed the tables form the results section and checked the guidelines.

Sincerely, Anna Nieboer